Author's response to reviews

Title: Prevalence and determinants of adolescent tobacco smoking in Addis Ababa, Ethiopia

Authors:

Emmanuel Rudatsikira (erudatsikira@llu.edu)
Abdurahman Abdo (abdo-abdurahmanabdo@yahoo.com)
Adamson S Muula (muula@email.unc.edu)

Version: 4 Date: 19 July 2007

Author's response to reviews:

Dear Editor,

We are pleased to submit a revised manuscript. The formatting changes that we were requested to do and have been done are listed below:

1. we have removed all underlining of headings and sub-headings
2. the abstract has the following sub-headings: Background, Methods, Results and Conclusions
3. We have updated the abstract in the submission to be the same as in the main text of manuscript
4. We have also declared that we do not have competing interests.

thank you editor.

adamson muula

Corresponding author